Author's response to reviews

Title: Management of acute renal colic in the UK: A questionnaire survey

Authors:

  Tunji Lasoye (Tj.Lasoye@kingsch.nhs.uk)
  Philip M Sedgwick (p.sedgwick@sghms.ac.uk)
  Nilay Patel (n1lay@hotmail.com)
  Chas Skinner (T.C.Skinner@soton.ac.uk)
  Nadeem Nayeem (nadeem.nayeem@uhl.nhs.uk)

Version: 4 Date: 3 December 2004

Dear sir/madam

Re: MS: 1882847153314930 - Management of acute renal colic in the UK: A questionnaire survey

Please enclosed details of the changes made to our manuscript following the referee's comments. The details have been structured in the same way as the referee's comments.

Thank you for your assistance.

Yours faithfully

Philip Sedgwick
Senior Lecturer in Medical Statistics,
Room 6.19, Level 6, Hunter Wing,
Department of Community Health Sciences,
St. George's Hospital Medical School,
London SW17 ORE.

General Comments
We have responded to all of the referee's general comments. We have indicated in the methods section under what circumstances the Chi-Squared test is valid, and when Fisher's Exact test would be used. A reference to this effect has also been included as suggested. We have also indicated which statistical software was used to perform the statistical analysis.

Major Compulsory Revisions
None were requested

Minor Essential Revisions
None were requested.

Discretionary Revisions
We have included a reference to the Kruskal-Wallis test as suggested. We do not wish to change the notation for the Kruskal-Wallis test statistic as suggested as it would be inappropriate to do so. However, the reference now provided gives an explanation as to why.